Author's response to reviews

Title: Posttraumatic stress symptoms and health-related quality of life: a two year follow up study of injury treated at the Emergency Department

Authors:

Juanita A Haagsma (j.haagsma@erasmusmc.nl)
Suzanne Polinder (s.polinder@erasmusmc.nl)
Miranda Olff (m.olf@amc.uva.nl)
Hidde Toet (h.toet@veiligheid.nl)
Gouke J Bonsel (g.bonsel@erasmusmc.nl)
Ed F van Beeck (e.vanbeeck@erasmusmc.nl)

Version: 4 Date: 20 December 2011

Author's response to reviews:

December 20th, 2011

Concerning – resubmission manuscript 2105678078526649
Posttraumatic stress symptoms and health-related quality of life: a two year follow up study of injury treated at the Emergency Department

Dear Editor,

We would like to thank you for accepting our manuscript entitled “Posttraumatic stress symptoms and health-related quality of life: a two year follow up study of injury treated at the Emergency Department” to be published in BMC Psychiatry under the condition that we revise the manuscript as recommended by The BMC Series Editorial Production Team.

We have added the statement that all authors read and approved the final Manuscript to the authors contribution paragraph.

We have included the country in the affiliation details.

We have removed all highlighting and tracking from the manuscript.

We have formatted the abbreviations section into a paragraph and moved it to the end of the manuscript (page 15 above the paragraph author’s contributions).

We have changed the titles "Introduction" and "Materials and methods" to "Background" and "Methods".

We have formatted the paragraph authors contributions into a paragraph.

We have deleted the title from the figure file and cropped the figure as closely as
possible.

We have checked your manuscript for any typographical errors.

The article is revised on the BMC Psychiatry web page. In case of further queries we are most willing to help.

Kind regards,

Juanita Haagsma
ErasmusMC
Department of Public Health
Rotterdam, The Netherlands